Author’s response to reviews

Title: Handling Health Promotion in Hospitals in diagnosis-related groups (DRGs): evaluation of systematic codes for health promotion activities in a multicentre project

Authors:

Hanne Tonnesen (ht02@bbh.hosp.dk)
Mette E Christensen (metschen@yahoo.dk)
Oliver Groene (ogr@es.euro.who.int)
Ann O’Riordan (info@ihph.ie)
Fabrizio Simonelli (f.simonelli@meyer.it)
Lagle Suurorg (lagle.suurorg@lastehaigla.ee)
Denise Morris (dmorris@uclan.ac.uk)
Peder Vibe (Peder.Vibe@skane.se)
Poul Erik Hansen (PEH@sst.dk)
Susan Himel (shimel@thc.on.ca)

Version: 5 Date: 18 July 2007

Author’s response to reviews: see over
Dear Editor of BMC Health Services Research

Thank you very much for your comments including only minor essential revisions, which have all been accepted and the text revised accordingly. Please find the revised manuscript uploaded.
The revisions are point by point scheduled below.
We are looking forward to hearing from you.

Best regards,
Majbritt Linneberg

On behalf of

Hanne Tønnesen
Director, WHO CC
Bispebjerg Hospital

<table>
<thead>
<tr>
<th>Remark/Reviewer 2</th>
<th>Comments and revisions</th>
</tr>
</thead>
<tbody>
<tr>
<td>On 3, ICD 10 was introduced before 1994. Authors may check WHO Classification</td>
<td>This has been checked and the manuscript revised.</td>
</tr>
<tr>
<td>Unit for the date of its introduction.</td>
<td></td>
</tr>
<tr>
<td>On page 3, ICF should be described using a full term when it is first used</td>
<td>The manuscript has been revised according to the comment.</td>
</tr>
<tr>
<td>instead of description 'ICF' on Page 4.</td>
<td></td>
</tr>
</tbody>
</table>